Author's response to reviews

Title: Salvia libanotica improves glycemia and serum lipid profile in rats fed a high fat diet

Authors:

Maya S Bassil (mbassil@lau.edu.lb)
Costantine F Daher (cdaher@lau.edu.lb)
Mohammad Mroueh (mmroueh@lau.edu.lb)
Nadine Zeeni (nadine.zeeni@lau.edu.lb)

Version: 6 Date: 13 October 2015

Author's response to reviews: see over
Dear Editor,

We were delighted to learn that our manuscript entitled “Salvia libanotica improves glycemia and serum lipid profile in rats fed a high fat diet” (by Maya Bassil et al.) was found to be potentially acceptable for publication in BMC Complementary and Alternative Medicine.

Please find below the answer to the editorial point that was sent to us. Kindly note that the change made on the revised manuscript is highlighted in yellow.

Kind regards,

Nadine Zeeni

Editorial point: Please confirm whether a voucher specimen of the plant material used in your study has been deposited in a publicly available herbarium, and include a statement to this effect in your Methods section. A deposition number should also be included, if available.

The following is now mentioned in the methods section of the manuscript (lines 91-92): “Also, a voucher specimen of the plant material used has been deposited at the herbarium of the Department of Natural Sciences, Lebanese American University (reference number: SL-2-2013).”